Author's response to reviews

Title: Excessive daytime sleepiness and falls among older men and women: cross-sectional examination of a population-based sample.

Authors:

Amie C Hayley (achayley@deakin.edu.au)
Lana J Williams (lanaw@barwonhealth.org.au)
Gerard A Kennedy (gerard.kennedy@cairnmillar.edu.au)
Kara L Holloway (khollo@barwonhealth.org.au)
Michael Berk (mikeb@barwonhealth.org.au)
Sharon L Brennan-Olsen (sharob@barwonhealth.org.au)
Julie A Pasco (juliep@barwonhealth.org.au)

Version: 6 Date: 15 May 2015

Author's response to reviews: see over
Dear Dr. Peeters,

Thank you for considering the further revised version of the original research article for publication in BMC Geriatrics.

We have addressed the typographical concerns raised by the editor. Please find attached to this letter our response to these comments.

We have uploaded a copy of the original manuscript marked with all the changes made during the revision process. The new text is underlined while the crossed-out text refers to the deleted original text. We have also uploaded a clean version of this final manuscript.

Thank you again for allowing us to resubmit a revised copy of the manuscript. We look forward to hearing from you in due time regarding our submission.

Many thanks,

Amie C. Hayley

IMPACT SRC, School of Medicine
Deakin University
C/- Barwon Health
PO Box 281, Geelong 3220, Australia
Tel: +61 3 5260 3564 Fax: +61 3 5246 5165
Email: achayley@deakin.edu.au
1. - Rename Introduction to Background.

This has now been amended.

2. - Include a heading 'Authors' Contributions' for the section designated.

This has now been amended.